The Price of Indoor Air Pollution: Evidence from Risk Maps and the Housing Market

Edward W. Pinchbeck^{*} Sefi Roth[†] Nikodem Szumilo[‡] Enrico Vanino[§]

Abstract

This paper uses the housing market to examine the costs of indoor air pollution. We focus on radon, a common indoor air pollutant which is the leading cause of lung cancer after smoking. For identification, we exploit a natural experiment whereby a risk map update in England induces exogenous variation in published pollution risk levels. We find a significant negative relationship between changes in published pollution risk levels and residential property prices. Interestingly, we do not find a symmetric effect for decreasing risk. We also show that the update of the risk map led higher socio-economic groups (SEGs) to move away from affected areas, attracting lower SEG residents via lower prices. Overall, our results demonstrate that indoor air quality has material economic effects on the housing market and provide novel policy-relevant insights into how the market responds to information on environmental risks.

Keywords: indoor air pollution; neighbourhood sorting; house prices; risk information; radon. JEL codes: Q53, H23, R21, R28.

^{*}University of Birmingham

[†]London School of Economics and IZA

 $^{^{\}ddagger} \mathrm{University}$ College London - Corresponding author: n.szumilo@ucl.ac.uk.

[§]University of Sheffield

[¶]Acknowledgments: We would like to thank David Rees and Marie Horton from Public Health England and Monika Szumilo from Imperial College London for their kind support. We also thank Gabriel Ahlfeldt, Johannes Boehm, Felipe Carozzi, Ludovica Gazzè, Matthew Gibson, Alastair Gray, Christian Hilber, Nicolai Kuminoff, Nico Pestel, Aurélien Saussay, three anonymous reviewers, and the editor, as well as seminar participants at the Center for Economic Performance (LSE), the Grantham Research Institute (LSE), University College London, the Hebrew University of Jerusalem, the University of Sheffield, the 2020 AERE Summer Conference, the 35th annual Congress of the European Economic Association and the 2020 Meeting of the Urban Economics Association for helpful comments and suggestions. This project uses Ordnance Survey AddressBase data which is Crown Copyright and is also based upon data provided by the British Geological Survey UKRI. All rights reserved. It also uses data supplied from Public Health England under the Open Government License.

1 Introduction

The adverse impacts of ambient pollution and its associated costs have received substantial public policy attention in the last few decades. Nevertheless, the health and economic consequences of indoor pollution are often overlooked. This is surprising given that the population in developed countries spends approximately 90% of their time indoors (Klepeis et al., 2001), and the documented substantial health implications from exposure to indoor pollution. More specifically, the World Health Organisation (WHO) estimates that indoor pollution is responsible for 2.7% of the global burden of disease and 3.8 million deaths every year. In Europe alone, it is estimated that a minimum of 99,000 deaths are attributed to indoor air pollution but experts suggest that the exact number is likely to be much higher (Holgate, 2017). Importantly, indoor pollution is not simply a by-product of ambient pollution, as there are numerous indoor sources of pollution including cooking, smoking and substances of natural origin such as radon. In fact, according to the US Environmental Protection Agency (EPA) the indoor concentrations of some pollutants are in many cases 2 to 5 times higher than those typical outdoor. As such, focusing on reducing ambient pollution concentrations without addressing the indoor environment would fail to provide adequate protection to the public against overall pollution exposure.

Assessing the economic effects of indoor pollution is challenging for several reasons, even compared to other non-market goods. First, large data samples on indoor air pollution are rarely available. Second, exposure to indoor pollution may be associated with unobserved factors that are also correlated with human health and well-being (e.g. income, diet and smoking). Finally, monetizing the total cost of exposure to indoor pollution is remarkably difficult, as exposure can lead to a wide range of observed or unobserved health and wellbeing costs.¹ This paper exploits a unique opportunity to overcome these challenges and examine the following research questions: What is the willingness to pay to avoid indoor air pollution? Which socio-economic groups are most affected by it? And what are the consequences of providing information to the public about indoor pollution risk?

We address these questions by looking at how an exogenous variation in the information provided by governmental sources about the level of indoor pollution risk is linked with variation in residential property prices in England. We focus on radon, an odourless, colourless and tasteless common indoor air pollutant which is formed by the natural decay of uranium from rocks and soil. Radon is considered to be the largest source of exposure to natural ionising radiation, which damages human lungs and causes lung cancer. Globally, exposure to radon at homes and workplaces is estimated to cause tens of thousands of deaths each year, making it the most important cause of lung cancer after smoking

¹Indoor pollution can affect a wide range of outcomes including mortality, morbidity, cognitive performance and educational outcomes (Duflo et al., 2008; Künn et al., 2019; Stafford, 2015; Gilraine, 2019; Roth, 2019).

(WHO, 2009; US-EPA, 2013). The main risk factors for getting lung cancer from radon depend on the level and duration of exposure, and whether the individual is a smoker or has smoked in the past. However, it is important to note that while smoking is an important risk factor, exposure to radon is also the number one cause of lung cancer among non-smokers. Due to increased awareness of these deleterious effects on health, radon exposure has become an increasingly important environmental concern for policymakers. Consequently, the US, EU and China have developed new radon policies for buildings and residents, and in many cases have commissioned radon maps of increasing accuracy and spatial granularity to increase awareness about this issue (Gruber et al., 2013; US-EPA, 2013; Zielinski et al., 2006). Nonetheless, there is very limited evidence on how effective those policies are in informing market participants about radon risk, and more generally about the economic cost of radon.²

We conduct our analysis using a unique data set that combines information on the universe of residential property transactions in England with detailed maps of radon risk provided by Public Health England that classifies radon risk into six categories of increasing risk, of which all but the first are classified as "radon affected areas".³ To overcome endogeneity concerns, we exploit a quasi-experimental radon risk variation resulting from the publication of an updated radon risk map in England in 2007, which did not incorporate changes in actual levels of radon, but simply new developments in risk modelling techniques. Our setting is also advantageous because since 2002 the UK standard conveyancing searches undertaken on behalf of house buyers must include information about radon risk based on the latest map.⁴ The combination of arguably exogenous variation and information disclosure to home buyers provides us with a unique opportunity to estimate the effects of pollution risk information in

 $^{^{2}}$ Our focus on radon is also advantageous from an empirical perspective. Unlike other common pollutants (such as particulate matter), outdoor concentration of radon is typically very low (as it dilutes quickly), and it is generally not consider as a health risk. This feature of radon is what makes it an ideal pollutant to use in our study, as it enables us to ensure that we estimate only the willingness to pay to avoid indoor air pollution and not ambient pollution or some combination of the two.

³The six categories of risk are based on the estimated percentage of houses in the grid square above the action level of 200 becquerels per cubic meter (Bq m^3): 0-1% (risk free area), 1-3%, 3-5%, 5-10%, 10-30% and >30%.

⁴These conveyancing searches are enquiries with public authorities that a specialist property lawyer or a licensed conveyancer makes to provide buyers with more information about the property that they plan to buy. These searches include enquiries that are sent to Local Authorities, which are the main units of local government in the UK. One part of the searches is the CON29 form: a standard set of questions set by the government, the Law Society and the Local Government Association. CON29 contains required and optional questions. Radon gas is on the required part, whereas flooding is on the optional part. The radon question asks if the property is in a "radon affected area", which is defined as any risk category higher than category one. There is no explicit requirement to report the specific category in which the property is, although the conveyancer can find this out from other sources. Importantly, conveyancing searches are required to obtain a mortgage. They are not required for cash buyers but will usually be undertaken. As a robustness test, we confirm that our main analysis yields similar results when using a sub-sample of properties with only mortgage buyers.

housing markets.

Our baseline specifications examine the effect of radon risk category changes on house prices using a repeat-sales approach. The principal measure of radon we use is the reclassification from the first category, i.e. reclassification into a "radon affected area". We find a highly significant negative relationship between changes in radon risk and changes in residential property prices. The results from our preferred specifications suggest that an upward reclassification of a property, from being in a radon risk-free category to being in a radon affected category, reduces property prices by about 0.8%. These results are economically significant and are similar to the estimated effects found in studies that measure the impact of flooding risk or earthquake risk on property prices (Bosker et al., 2018; Naoi et al., 2009). In contrast, we find no evidence that downwards reclassifications of radon risk (i.e. lower radon risk) affect house prices. A possible explanation for these asymmetric results might be the fact that buyers are responsible for conducting the conveyancing search, which includes information about the current radon risk level, and that sellers might not be informed of the latest radon risk reclassification of their area. However, we are unable to make strong claims about this because there are substantially less data points for which radon risk levels fall.

We next test whether radon risk induces sorting. We find that it does, and more specifically we show that the share of higher educated, home owning and higher social status residents decreases in response to upward radon risk reclassification. Once again, we find no effect of downward reclassification. These findings have two important implications. First, they suggest that radon risk disproportionately affects lower socio-economic groups in our society, creating a source of environmental injustice, in line with previous theoretical predictions (Hausman and Stolper, 2021). Second, it means that our previous estimates are "local", in the sense that they reveal the Willingness To Pay (WTP) of the specific buyers who are purchasing houses affected by radon and not the average buyer (see Greenstone (2017), Gamper-Rabindran and Timmins (2013) and Davis (2004) for more details). Therefore, we propose and apply a new theoretical framework, following Bayer et al. (2007) and utilising results from Besley et al. (2014), that uses transaction probabilities to account for sorting in order to calculate the average WTP. We also present several robustness tests and placebo exercises that provide further support to our empirical analysis and validate our estimates.

Our study provides several important contributions to the literature and to policy-making more broadly. First, despite the growing body of work on the capitalization of ambient air pollution (Chay and Greenstone, 2005; Currie et al., 2015), the paucity of direct measurements of indoor air pollution mean we are among the first to provide an hedonic estimate of the economic impact of indoor air

pollution. To the best of our knowledge, the only other revealed-preference estimates of WTP for indoor air quality improvements are inferred from the demand for air purifiers in China (Ito and Zhang, 2020). The authors of this study develop a novel framework to estimate WTP for indoor air quality improvements from information on the air purifier's effectiveness in reducing indoor particulate matter and the price elasticity of demand. Although we lack home-specific measures of indoor air pollution, the risk categories in our data are arguably a very close approximation to this ideal, because they are derived from the underlying measurements and because they are very granular. Furthermore, even though radon is linked with significant health risks by the epidemiological literature, it has not yet received much attention in economics and our paper addresses this gap as well. In fact, the only two exceptions are a paper by Smith and Johnson (1988) which studied how households form risk perceptions using a survey of households' responses to information about risks associated with radon in Maine, and a paper by Howe and Shreedhar (2020) that evaluated if a 'Healthy Home Checklist' postcard nudges households to purchase a radon test in Ottawa.

Second, consistent with approaches in other countries and other domestic policies (e.g. on energy efficiency), the UK Government radon policy centres on increasing information availability to allow markets to price environmental risks. Our paper provides critical insights into how the market responds to this information, complementing a growing literature on the effects of access to pollution information that has so far solely focused on ambient pollution (Cutter and Neidell, 2009; Zivin and Neidell, 2009; Barwick et al., 2019), and to the empirical literature on the role of information on environmental risks and on consumer choices more broadly (Hastings and Weinstein, 2008; Jessoe and Rapson, 2014; Wichman, 2017; Bakkensen and Barrage, 2017; Bakkensen and Ma, 2020). Importantly, our distinctive setting enables us to purely estimate the effect of information shock as the underlying risk (actual pollution level) has not changed. This is unique because most papers in the literature study an exogenous shock to the amenity itself (e.g. flood risk) and the price or other responses to this change are a composite of the change in the amenity itself and its salience.

Third, our work also speaks to the literature on environmental justice, which denotes various channels through which environmental costs and poverty or race may be correlated, including household sorting (or "coming to the noise/nuisance") and selective siting of polluting facilities (Banzhaf et al., 2019; Hausman and Stolper, 2021). The evidence on the extent to which sorting drives the observed correlations in this literature is mixed, in part reflecting considerable empirical challenges (Banzhaf et al., 2019). We contribute to this literature by showing that different socio-economic groups sort in response to radon risk change.

Fourth, our study contributes to the existing hedonic literature on air pollution by overcoming two important and common empirical limitations. More specifically, the institutional set up we study determines that information on property-level radon risk is made available to buyers through standard conveyancing searches. This addresses an important concern regarding prior research studying the impact of air pollution on property prices, which are less certain about market participants awareness of pollution levels. Recent evidence on the effects of environmental information disclosure on house prices and other outcomes suggests this is an important concern (e.g Pope, 2008; Moulton et al., 2018). Additionally, we propose and utilise a new theoretical framework to estimate the average willingness to pay in the presence of sorting.

Finally, our paper demonstrates that the stakes are high for policymakers working on indoor air pollution. Due to its geology, the UK has relatively low levels of radon compared to those found in other countries, notably the US and China.⁵ Despite this, our estimates imply that if unmitigated by remedial measures the presence of radon in our setting would reduce the total market value of residential properties in England by as much as $\pounds 5.8$ billion.⁶

The rest of the paper is structured as follows: the next section provides background information about radon risk and risk information policy in the UK; section 3 presents the data and our study design; section 4 discusses the empirical strategy; section 5 presents the main results of our analysis; section 6 proposes and utilises a new theoretical framework to account for sorting and estimate the average WTP, while section 7 concludes.

2 Background on radon and the UK

2.1 Background on radon

Radon is a colourless, odourless and tasteless gas which is formed by the natural decay of uranium that occurs in all rocks and soil. Once escaped from the ground into the air, it decays and produces radioactive particles which tend to concentrate in enclosed spaces. The amount of radon in houses, schools and workplaces depends on a variety of factors including the amount of uranium in the underlying rocks and soil, the available routes for radon to enter the building, and the rate of exchange

⁵According to the EPA, approximately 1 out of every 15 houses in the US have elevated levels of radon.

⁶This range is obtained by multiplying the average impact of being reclassified to a radon affected area (0.8%) by the number and average value of all houses in the housing stock classified to be in radon affected areas according to the latest and most accurate radon maps available (25x25 meter grid).

between indoor and outdoor air (WHO, 2009). Radon is usually measured in becquerels per cubic meter (Bq m^3) or in picocuries per liter (pCi/L). In the UK and in many other countries, a radon level of 200 (Bq m^3) or above is classified as "Action Level", which means that householders are advised to take actions to reduce radon through remediation works (e.g. by installing a radon sump). The average radon level in UK houses is 20 (Bq m^3), but there is significant variation across geographical areas and some dwellings have yielded readings of more than ten thousand Bq m^3 (PHE, 2018).

According to the medical literature, radon can affect the human body as radioactive elements are inhaled and enter the lungs where they emit radiation. This radiation (mainly alpha particles) is absorbed by nearby tissues which can lead to lung cancer. Early epidemiological studies evaluated this link empirically among miners, and established that exposure to high levels of radon is associated with an increased risk of lung cancer (Radford and Renard, 1984; Howe et al., 1986; Tirmarche et al., 1993). Later studies have examined the relationship between residential exposure to radon and lung cancer in the general population. This strand of the literature, which relies on pooled large scale epidemiological studies, provides robust evidence that residential exposure to radon is linked with a significant number of lung cancers in China, North America and Europe (Lubin et al., 2004; Krewski et al., 2005; Darby et al., 2006). While there is a small theoretical risk that radon exposure can also cause cancer to other organs, there is currently no strong empirical evidence to support this hypothesis (AGIR, 2009).

Approximately 1,100 lung cancer deaths a year in the UK and 21,000 in the US are attributed to exposure to elevated levels of radon. Although these numbers are considerably lower than deaths attributable to smoking, radon is the most important cause of lung cancer after smoking (US-EPA, 2013; Grav et al., 2009). The main risk factors for getting lung cancer from radon depend on the levels of radon exposure and whether the individual is a smoker or has smoked in the past. For example, if 1,000 people who smoke were exposed to a radon level of around 0.5 Bq m^3 over a life time, about 150 of them could get cancer compared to 36 people among a group of 1,000 people who have never smoked and were exposed to the same level of radon. Nevertheless, it is important to clarify that non-smokers are still at significant risk from radon exposure, and according to the EPA radon is the number one cause of lung cancer among non-smokers. There are several methods for radon prevention and mitigation at dwellings, such as increasing under-floor ventilation, installing radon sump systems, and sealing floors and walls. The appropriate remedial strategy depends on several factors including radon levels, sources and the type of floor. Importantly, none of the techniques can guarantee complete eradication of radon, and often a combination of strategies is required to reduce the risk. This is important given the fact that measuring radon concentration is a complex process, as readings can vary between rooms, over times of day, months of the year or even with phases of the moon. Costs

of these remedial techniques vary significantly, but as a rough guideline the approximate costs range between £200 and £2000 according to Public Health England. However, in practice a high proportion of affected properties, both in the UK and in other countries, do not have any remediation measure in place (Zhang et al., 2011). This could be due to non-monetary costs of installation (hassle costs), budget constraints for lower socio-economic groups, or low awareness of the health risks related to radon.

2.2 History of radon information and policy in the UK

Developments in understanding the health risks of ionizing radiation after the Second World War led to investigations of radon levels in UK houses in the 1970s and 1980s. These initial local measurement schemes developed into a systematic measurement program which tried to identify areas and houses exposed to the risk. The concept of "radon affected areas" was introduced in the 1990s and defined as areas where 1% of houses were above a pre-specified action level. In 1993 several parts of the UK were designated as affected areas and in 1996 the first complete nation-wide map of radon was published. The map was based on over 250,000 measurements, and denoted radon risk in 5km grid squares using five categories of risk based on the percentage of houses in the grid above the action level: 0-1% (risk free area), 1-3%, 3-10%,10-30% and >30%. The measurements in each dwelling were made with two detectors over a period of three months and were calculated using temperature corrections to enable to produce accurate annual estimates.⁷

A second radon map was issued in November 2002 and was in use from 2003. This map (the "2003 map") updated the 1996 map based on additional measurements (over 400,000). It also included an additional risk category, as the 3-10% category was split into 3-5% and 5-10%. Additional measurements allowed mapping the affected areas of the South West of England in a $1km^2$ grid. A third iteration of the radon map was then released in November 2007 (the "2007 map"), as shown in Figure 1 presenting the evolution of these maps. This updated map incorporated further improvements to probability models, which now included not only measurements but also geological information as input variables (Miles et al., 2007).⁸ Importantly, two versions of this 2007 map were created from

⁷Houses included in the measurement program were selected based on progressive measurements of areas where initial reading were high and on geological data.

⁸Specifically Miles et al. (2007) state that, "The previous maps published separately by the NRPB and BGS grouped domestic radon results either by grid square or by geological unit, before applying lognormal modelling. Both of these mapping methods ignore some part of the geographical variation in radon potential: grid square mapping ignores variation between geological units within grid squares, and geological mapping ignores variation within areas sharing combinations of geological characteristics. It was realised by the HPA and BGS that combining the two methods could

the same underlying data. The first (the "25m 2007 map") was a map using a 25x25 meter grid for the whole of England. Unlike the earlier maps, this version was not available to the public or to local government organisations free of charge, but required a fee to be accessed (a paid subscription for institutions including government organisations). However, a second publicly available version of the 2007 update was released, which indicated radon risk in 1km grid squares (the "1km 2007 map"). This $1km^2$ map was created by taking the maximum category of radon risk in all the 25 meter squares within each 1km square.⁹

Radon information can be obtained by individuals in two main ways. First, maps are available online for download, so potential buyers could in theory find this information themselves. However, since it is unlikely that the average buyer would actively search for such information online¹⁰ we believe (and later document) that the main source of radon information that buyers receive is via the set of questions submitted to Local Authorities (LAs) by specialist property lawyers or licensed conveyancers on behalf of their clients when a property is traded or a development application is made.¹¹ These are known as Local Authority searches and the specific form used to make enquiries is known as CON29. CON29 contains a standard set of questions which are jointly agreed by the government and the Law Society. Part of the form indicates required questions (CON29r) and another part contains optional questions (CON290). Information about radon has been a part of this process since 1994 and it was made compulsory in July 2002 (it was previously optional). Conducting Local Authority searches is a formal requirement for mortgage lenders, so we can be sure that all mortgaged buyers receive radon information (representing around 70% of all transaction in the UK according to the ONS). While cashbuyers are not required to do so, we expect most will as the searches are inexpensive and solicitors will recommend them. The information about radon is based on risk categories, but the CON29 is only strictly required to state if a property is in a radon affected area (any category other than category 1) and not its actual risk category.¹² Buyers obtain CON29 information only at a relatively late stage in the buying process, since the LA searches are performed after an offer is accepted, but before final sales price is agreed and the deposit (usually 10% of the final sales price) is paid. Frequently, the give more accurate mapping than either separately. The two organisations cooperated to develop a joint geological/grid square mapping method. This atlas is the outcome of applying this method to radon mapping in England and Wales, and supersedes the previous atlas of radon potential in England and Wales".

⁹A comparative example of these two set of maps for the Sheffield Local Authority District is provided in Figure A1 in the Appendix.

¹⁰Analytics from Google Trend show a limited and stable number of online searches in the UK about radon maps through this period, even around the release of updated maps. Also see section 5 for more analysis on this subject.

¹¹Local Authorities are the main unit of local government in the UK.

¹²Specimen of a standard CON29 form could be found on the UK Law Society website: https://www.lawsociety. org.uk/en/topics/property/con29-forms. Figure A2 in the Appendix shows an example of the demand and answer about radon provided in a standard CON29 form.

final sales price differs from the accepted offer to reflect information disclosed to the buyer about previously unknown characteristics of the property (lack of required documents, environmental hazards, revealed extensive work required, presence of Japanese knotweed, etc.), as in England an offer does not become legally binding until contracts are exchanged. In addition, the results of the conveyancing searches are not transmitted to the sellers, who might be informed about their outcomes only by buyers.

The transmission of information contained in the radon maps to home buyers is an important consideration in our empirical work. In late 2002 copies of the 2003 map were provided to all Local Authorities (LAs) in the country, and contemporaneous guidance issued by the National Radon Protection Board (NRPB) indicated that the map should be used by LAs to answer enquiries by conveyancers about whether houses were in radon affected areas. For the 2007 map update, things are more complicated as Local Authorities had a choice about which map to use. Around half of the Local Authorities in the country have elected to licence the more spatially detailed 25x25 meter version of the map for part or all of the period since it was released, while the other half have elected to use the freely available $1km^2$ version of the 2007 map. In our empirical work we will define house level radon information based on the version of the 2007 map the associated LA had at the time of the house sale, and further exploit the two map versions for robustness checks.¹³

3 Data and Study Design

Our final dataset combines information on house sales, radon risk and various socioeconomic characteristics from several administrative sources. Data on house sales come from the Land Registry, which contains all housing transactions in England between the years 1995-2018. From this source,

¹³Besides the radon maps and the CON29 form, there are two further institutional details which are somewhat related. First, newly built houses are subject to radon policies. The general policy since 1990 has been that houses with radon levels above $200Bqm^3$ are advised to apply preventive measures and all new buildings in affected areas have to apply these measures (National Radiological Protection Board, 1996). Second, a Home Information Packs (HIP) policy, in place from late 2007 to May 2010, required house sellers or their representatives to disclose a number of documents and pieces of information to buyers about the house before a sale, including an Energy Performance Certificate, local authority searches, title documents, guarantees, etc, as well as to state if the house is in a radon affected area as identified by the Health Protection Agency. While the HIP policy could signal a mechanism for sellers to learn of radon risk changes to their house, in practice this assumption could be tenuous as it would require that (a) the HIP contains accurate and up to date radon information from a new LA radon search, and (b) the seller carefully checking the HIP radon question. The first condition is undermined because the regulations do not specify which map should be used, and by evidence from a survey that found five out of six HIPs contained "inaccurate, incomplete or missing information" ("Home Information Packs: a short history", House of Commons Research Paper 10/69). The second also seems improbable given that the vast majority of sellers would be unaware of the 2007 radon map update (or even that radon maps are periodically updated), and hence unlikely to carefully check through a 100 page document for this information.

we take over 2.4 million transactions that took place during our 2003-2011 study period. This is the population of all repeated transactions of the same property which is a sub-sample of the population of all 6.48 million transactions in England in this period. Our data on radon risk comes from radon maps provided by Public Health England (PHE) which are the publicly available maps and the 25 meter square map provided by the British Geological Survey. The public maps are presented in Figure 1. Notably, the three maps include revisions of the radon level in numerous locations with the vast majority of revisions increasing the level of radon risk (see Table 1).

In order to focus our study on areas where we can be certain that buyers received information about radon and what map was used to provide it, we only consider transactions in which the first sale occurred between January 2003 and October 2007 and the second sale between 2008-2011.¹⁴ The full database consists of 601,6554 pairs of transactions, as shown in Table 1, where for each property we observe two transactions, one before the map update and one after.

There are 6 radon categories in each map, and house level changes can theoretically be in any of the 36 possible combinations. However, as shown in Table 1, in practice about 72% of dwellings in our sample stay in the same category (n=435,090). Of those that do move categories (n=166,565), more than 64% start in category 1 and move up, while only 13% start in a category higher than 1 and move down one or several categories. Moreover, some combinations of starting and ending category are very thin, especially when we condition on area fixed effects. We therefore collapse the dimensionality of this problem by focusing on particular start/end category combinations that are well-populated, notably moving to and from a radon affected classification (i.e. categories above 1), as shown in Figure 2 where we map the spatial distribution of areas reclassified as radon-affected, those reclassified as being radon-free, and those unaffected by the reclassification between the 2003 and the 2007 maps. This strategy also better reflects the information provided to buyers about properties radon-risk through the CON29 forms. To create our final sales dataset, we first geocode each property using the AddressBase database from the Ordnance Survey. This gives us the exact location of each house, which enables us to assign a radon risk category to each transaction from the radon risk maps. By using ancillary data which tell us the periods when Local Authorities had a subscription to the 25x25 meter grid version of the 2007 radon map, we ensure the radon risk in our data matches the radon information buyers receive from the CON29 form.¹⁵ Note however that where CON29 information comes from the $1km^2$

¹⁴Starting our sample period in 2003 implies that we use two iterations of radon maps and we focus only on sales when radon questions were compulsory in the CON29 form. With the most recent radon map issued in November 2007, we drop the final 2 months of 2007 because it is unclear which map was used. We choose to end the sample period in 2011 as this is the last year for which we have rich demographic information from the Population Census.

¹⁵We obtained this information about map usage from The British Geological Survey.

map, individuals can purchase information from the 25 meter square map which may indicate a lower, but not higher, radon category (the $1km^2$ map gives the local maximum of the 25x25 meter tiles). This implies we may sometimes have "false positive" radon affected classifications in the $1km^2$ map.

Besides this property-level dataset, we also create several area level datasets. For balancing checks and to test for sorting, we obtain micro-neighbourhood information on socioeconomic characteristics such as education, age and ethnicity from the 1991, 2001, and 2011 Censuses. The spatial units we use are Output Area (OA), the smallest census geography in the UK, with an average of around 100 residents.¹⁶ We measure the radon level in each area by assigning the median radon risk for all houses in the same neighbourhood. Finally, to examine the effect of radon on transactions, we also create two panel datasets for the period 2003-2011 at the level of (i) the radon map grid square tile and (ii) the Output Area.¹⁷

Table 2 presents summary statistics for all property transactions in our dataset. The first two columns, considering all transactions, show that cross-sectionally transaction prices in radon affected areas do not appear to be statistically different from those in radon-free areas, although the mean house price is slightly lower for transactions in categories higher than one. The next three columns consider only repeated sale transactions, showing that prices of properties where radon did not change grew faster than in places where radon risk changed (either upwards or downwards). Figure 3 provides another insight into the evolution of prices, comparing properties that were reclassified from radon free to radon-affected areas with unaffected properties before and after the reclassification. As evident from the figure, reclassified properties experienced a sizable reduction in price growth following the reclassification in October 2007, compered to properties which were not reclassified. Naturally, these comparisons are very general as they do not account for confounding factors. In the next section we provide detailed explanation of possible confounders and how our empirical strategy overcomes them.

4 Empirical Strategy

Following the standard hedonic approach to modelling house prices, we begin with the following simple equation:

$$ln(P_{it}) = \beta R_{it} + \rho X_i + \epsilon_{it} \tag{1}$$

¹⁶OAs are the smallest of a hierarchy of nested Census geographies. They sit within Lower Layer Super Output Areas (LSOAs) which in turn are nested within Middle Layer Super Output Areas (MSOAs).

¹⁷More information provided in Appendix A.2.

Where P_{it} denotes the transaction price of property *i* in year *t*, R_{it} is the level of radon risk category for the transacted property given by the map used by the Local Authority at the time of the transaction, X_i is a vector of time invariant property characteristics, and ϵ is an idiosyncratic error term.

The usual identification problem is that radon risk levels can be correlated with both the features of the house and with property price. For example, geological conditions could affect the level of radon risk but also the amenity value of the location or construction materials used to build houses. As such, estimating equation 1 would yield a biased estimate in the presence of unobserved correlated factors. To overcome this problem we take advantage of the popular approach in the real estate literature and adopt a repeat-sales approach (Cannaday et al., 2005). More specifically, if the same property is transacted also in year τ (where $\tau > t$), we can write a first-difference model of the following form:

$$ln\left(\frac{P_{i\tau}}{P_{it}}\right) = ln(P_{i\tau}) - ln(P_{it}) = \beta \Delta R_{i\tau t} + \epsilon_{i\tau t}$$
⁽²⁾

The repeat-sales approach removes all time-invariant characteristics and focuses our identification on changes in radon risk levels within the same property over time. The approach is particularly compelling in our context because geological conditions and actual radon levels do not change over time, yet the estimates of the radon risk do due to map updates. As radon only affects humans through the impact on their health, β captures the impact of the health risk associated with radon on prices. Since time varying unobserved correlated variables remain a potential threat for causal identification, we allow for house prices to change over time by including a fixed effect θ for the first and second years, τ and t, and quarters, κ and q, which effectively strips out any time effects in log price changes and seasonal price effects:

$$ln\left(\frac{P_{i\tau}}{P_{it}}\right) = ln(P_{i\tau}) - ln(P_{it}) = \beta \Delta R_{i\tau t} + \theta_t + \theta_\tau + \theta_q + \theta_\kappa + \epsilon_{i\tau t}$$
(3)

Furthermore, we also adjust for local housing cycles by allowing each area to have its own trend in prices between periods τ and t. We define areas in one of two ways, one based on administrative geography and the other using an arbitrary spatial grid. For the former, our units are Census Middle-Layer Super Output Areas (MSOAs), which have populations of between 7,000 and 10,000 people. For the latter, we group homes together by rounding the centroid of the full postcode - which typically contain between 15 and 100 homes - to a 1km grid. This postcode grid has the advantage of yielding a tight spatial grouping that does not fully overlap with the radon map grid or administrative units and is used in our main estimates. Finally, we separate areas according to which version of the 2007 map (25x25 meter or $1km^2$) was in use by the Local Authority at the time of the second sale, which

effectively means our identifying variation comes from comparing sales of homes that become radon affected or not under the same version of the map.¹⁸ Specifically, our main specification takes the following form:

$$ln\left(\frac{P_{i\tau}}{P_{it}}\right) = ln(P_{i\tau}) - ln(P_{it}) = \beta \Delta R_{i\tau t} + \theta_{rt} + \theta_{r\tau\mu} + \theta_q + \theta_\kappa + \epsilon_{i\tau t}$$
(4)

Where θ_{rt} and $\theta_{r\tau\mu}$ interact year of sale and area r fixed effects (postcode grid or MSOA), and in the latter case with an indicator which denotes whether the parent Local Authority had a contract to use the 25 meter square map. Therefore, our identification is based on changes in radon risk levels within the same property over time after accounting for all time-invariant characteristics of the property and factors common to the local area.

Importantly, for equation 4 to reveal a causal estimate we have to assume that without the treatment changes in house prices in the treatment group would have followed the same trend as in the control group. We test this assumption in two ways, first we assess if the two groups followed the same trends before the treatment. In Fig. 3, price indices of the two groups are plotted against time and clearly follow the same trends before 2008. Second, in section 5.3 we test the parallel pre-trend assumption more formally and find no evidence that it is violated. Besides this, we can also evaluate whether the main treatment in our analysis (radon level going from "risk free" to "radon affected" relative to remaining "risk free") is correlated with potential determinants of house prices. Only if our treatment is plausibly uncorrelated to these factors, we can use our estimated price effects to calculate the average willingness to pay (see section A3 in the Appendix). To do so, we obtain data for micro neighbourhood level characteristics from the 2001 and 1991 Censuses and regress these on an indicator for radon level going from "risk free" to "radon affected" under the 2007 map change, together with area fixed effects and some time invariant spatial controls, restricting the attention to places that were classified as being radon risk free in the initial period. Our regressions take the form:

$$y_j = \gamma R_j^+ + \theta_{r\mu} + \rho X_j + \epsilon_j \tag{5}$$

where y_j represents a share variables for 2001, or a change in the share variables between 1991 and

¹⁸Note that the radon map version is not collinear with the fixed effects, as map use changes over time and can also vary within postcode grids. As the $1km^2$ map reflects a local maximum, radon designations may have different implications under the different maps; these controls thus ensure we are comparing like with like. These controls also mitigate effects of false positive radon classifications in the $1km^2$ map which arise because people can purchase the 25 meter square map. False positives may be particularly prevalent in postcode grid cells where both maps are used and, as conditional treatment variance is higher in the 25 meter square map, ignoring the map version tends to increase the weight our estimator puts on these false positives. Our results are materially unchanged if we instead drop all observations in postcode grid-year cells that contain both version of the map in which case this control is obsolete.

2001 for Census Output Areas (OA) j, while R_j^+ is an indicator for (median) radon level going from "risk free" to "radon affected" under the 2007 map change, $\theta_{r\mu}$ is an interaction between the area rfixed effect and an indicator μ denoting whether the associated Local Authority had a contract to use the 25 meter square map at any time after the 2007 map was released, and X_j includes distances to the labour market centre, the nearest coast, and nearest rail station. The additional controls capture trends correlated to geology but also to long term gentrification (they drop out of equation 4 as they are time-invariant).

In the first column of Table A2 we find that most area characteristics are uncorrelated with our radon measure when we use our preferred $1km^2$ postcode grid local area fixed effects. There are some exceptions to this. In particular, we find that the share of workers in high income professional and managerial occupations is correlated to radon, although the magnitude of the effect is small. The same holds true for education (which is unsurprising given that this is highly correlated with the occupation measure ($\rho = 0.83$)), age and ethnicity. However, this is not necessarily a concern, since our main identification strategy eliminates time-invariant sources of bias. Reassuringly, in the second column we find no significant correlations between radon risk increases and changes in the share variables between 1991 and 2001 for characteristics that are available in both Censuses, except for age and ethnicity but the magnitudes of these coefficients are very small. Moreover, later in the paper, we test for the presence of price trends correlated to our treatment locations (using periods in which there were no changes in radon maps) and find no evidence of trends correlated to our treatment. Besides this, we also test whether radon risk is correlated with ambient air pollution which has been shown to affect house prices in numerous studies (e.g. Chay and Greenstone (2005)) and can therefore pose a risk for causal interpretation. More specifically, we examine the correlation of OA level ambient air pollution measured by PM10 and OA level radon risk.¹⁹ In Table A1 columns (1) and (2) we regress levels of PM10 on levels of radon risk in 2003 and 2008 with TTWA fixed effects. Column (3) repeats this for changes in both measures of air pollution. The results are statistically insignificant and the coefficients are very small, indicating that ambient air pollution is not correlated with either levels or changes in radon risk in our data. In addition, later on in Table 5 we test if any other possible local trends could be correlated with changes in radon, by conducting a placebo test examining the impact of the change in radon maps introduced in 2007 on transaction pairs in which both first and second sales happen before the publication of the 2007 map. In line with previous tests, we find no effect,

¹⁹The information on ambient air pollution (PM10) comes from the Department for Environment Food and Rural Affairs (DEFRA). To generate PM10 measures we keep stations that have less than 10% missing values for these years and we assign pollution level to OA by linking it with the three closest monitoring stations to each OA centroid. We then use the mean value of those three measurements weighted by the inverse squared distance between the monitoring station and the OA centroid.

corroborating once more the random distribution of our treatment.

To claim that the price effect reflects the willingness to pay to avoid radon we require several additional assumptions. First, housing market participants (buyers and sellers) must be aware of the radon risk and take it into account when making buying and selling decisions. For example, if buyers are unaware of higher radon risk in a property, they will not ask for a discount and our estimates should be zero. While this fundamental assumption is being made implicitly in all papers that rely on the hedonic technique, it is not frequently discussed or addressed explicitly.²⁰ We believe that our unique context enables us to ease concerns regarding a violation of this crucial assumption as buyers receive information about radon risk levels when buying properties. More specifically, buyers receive information on radon risk levels as it is included in the conveyancing process through the local authority search (CON29), which is based on the same radon risk data that we use in our study. Interestingly, while the owner should have known what the radon risk was at the time of the first transaction, they will not necessarily be aware of updates to radon risk in their area. Therefore, if new data reveal that the radon risk in an area is reduced, a rational buyer might not pass this information to the seller and we would not expect to see an effect on property prices, as information about the map updates is not widely advertised and sellers do not typically pay for conveyancing searches ahead of sales.

Second, in order to claim that the estimated effect is the price households attach to indoor air pollution risk, we have to measure changes in air pollution risk observed by households. One possible concern is that radon remedial measures may introduce a misalignment between the level of risk observed in our data and by buyers, since we cannot observe if properties have remedial measures in place. For example, a change in the level of risk which in our data is between categories 2 and 6 could be less consequential for the buyer if the house is equipped with radon mitigation devices. Mindful of this issue, in the majority of specifications we will focus on properties for which the first transaction occurred when the contemporary map indicated no risk of radon. These houses had no, or at least very small, incentives to install radon mitigation devices before the update of the map. Furthermore, in our downwards movements analysis we exclude properties that were new at the first sale, as newly-built houses in radon-affected areas are advised to install radon protection during construction, and we cannot observe whether they did and what these are. Another possible source of measurement error is that owners can measure the actual level of radon in their houses and may thus have different information than reflected on the map. For example, if the seller tests the house that has been reclassified into a high risk area and the test shows no pollution, our measure of risk would be biased. We will address

 $^{^{20}}$ There are several exceptions to this, including early work by Kask and Maani (1992). See Pope (2008) for a recent example of an environmental application that examines the effect of information disclosure.

this concern in Section 5.3.

Third, we must also be assured that the institutional features of our setting do not potentially drive a wedge between our estimates and the value that households truly place on radon risk. We address various concerns, including those relating to inaccuracy of radon information in our later robustness checks.

Finally, it is important to note that the hedonic model only reveals the marginal willingness to pay (WTP) of the marginal buyer. This means that, in order to interpret our "direct" hedonic estimates as the average WTP of the population, we would have to make a number of assumptions, including that there is no sorting based on radon (Greenstone, 2017; Ekeland et al., 2002; Gamper-Rabindran and Timmins, 2013; Davis, 2004). In the next sections we will test this assumption directly and in Section 6 we will propose and apply a new theoretical framework to estimate this policy relevant parameter in a context with sorting.

While we obtain the impact on house prices from our central empirical approach above, we also conduct a number of ancillary regressions. To estimate the effect of radon risk on sorting, we utilise the following cross-sectional specification:

$$y_j = \gamma_1 R_j^+ + \gamma_2 R_j^- + \theta_{r\mu} + \theta_{\eta} + \rho X_j + \epsilon_j \tag{6}$$

This is similar to the balancing estimation described in equation 5, albeit here the dependent variables are the change in shares between the 2001 and 2011 Censuses. Moreover, because here we wish to examine the effect of moving to and from a radon affected classification, we add places that move down to a radon-free classification in our sample and include fixed-effect for the initial 2003 radon category (θ_{η}). This implies that estimates can be interpreted as the average change in the outcome when radon changes, relative to remaining in the same initial radon category.

5 Results

5.1 Main Results

We begin our empirical investigation by examining whether upwards and downwards movements in house level radon risk categories affect property prices using first difference estimations.²¹ In all cases

 $^{^{21}}$ For clarity of exposition we present upwards and downwards changes in separate regressions but collapsing them into a single regression yields effectively identical results.

we control for whether a house was new at the previous sale, and cluster standard errors on the radon grid square, which is the spatial scale at which radon varies. Recall that radon category 1 is denoted as a "radon risk free area" in the local authority search (CON29), whereas all other categories are denoted as being in "an area that is affected by radon". Commensurate with the institutional set up, Table 3 examines whether moving to and from category 1 affects house prices. In the first two columns of this table the fixed effects are derived from the 1km postcode grid, and in the last two MSOAs. These specifications control for changes in local amenities such as school quality, public transport or ambient air pollution, but also for changes in the composition of the neighbourhood (i.e. sorting), which helps identify the direct impact of map changes on affected houses (Ekeland et al., 2002).

In panel A, we test for upwards risk reclassifications, using samples composed of sales that stay in category 1 (the baseline category), and houses that move from category 1 to other, by definition higher, categories. In column 1, we examine moves from category 1 (risk free) to the lowest category classified as affected by radon (category 2). Despite the relatively small increase in radon risk between categories 1 and 2, this upward risk reclassification decreases house prices by 0.86%. In column 2 we expand the sample to include properties that start in category 1 but move to higher risk categories. The estimated parameter is slightly smaller in magnitude, but statistically indistinguishable from our estimate in column $1.^{22}$

Given that the updating of the maps led to reclassifications of properties downwards as well as upwards, we test downwards moves in panel B of Table 3. Our approach relies on comparing houses that stay in a given category with others that move from the same category to category 1. However, we acknowledge that our estimation sample is much smaller than for upwards moves, which limits the strength of the conclusions we can draw. In column 1, we examine the effect of moving from category 2 to category 1 relative to staying in category 2. We obtain a small positive coefficient that is not significantly different from zero. In column 2 we expand the sample to any house that moved to category 1 or stayed in the same radon-affected category, identifying the average downward effect by adding the baseline radon category into the interactive fixed effects. Again, the estimate is statistically insignificant and the coefficient is even smaller, implying that conditional on our 1km postcode grid trends in prices, a house reclassified from a radon affected to a radon-free area does not command a statistically significant price premium.

 $^{^{22}}$ As we discuss further below, it may be that the smaller magnitude of the coefficients when we add higher risk categories follows from unobserved radon remedial works in higher risk category houses. In many later regressions we will focus solely on properties that move from category 1 to category 2 for this reason.

This suggests an asymmetry in price responses to radon risk, which is consistent with buyers and sellers acting on information about higher radon risk but not acting on information about lower radon risk. One possible explanation for this is the asymmetry in radon information between buyers and sellers. As mentioned above, buyers receive radon risk information from conveyancing searches, but sellers will not necessarily be fully aware of the latest radon risk classification in their area, as map updates are not widely advertised and sellers do not typically pay for conveyancing searches ahead of sales.²³ Hence, with information mostly coming from the buyer-side, only upwards movements in radon risk classification get priced in. Nonetheless, we acknowledge the limitations imposed by our data here and moreover that there are other possible mechanisms that could lead to this finding. For example, a persistent sorting effect or a stigma effect could lead to similar results (McCluskey and Rausser, 2003; Ioannides and Zabel, 2003).

In columns 3 and 4 of Table 3, we use MSOA rather than the 1km grid fixed-effects. Reassuringly, the estimates remain similar to the findings in the first 2 columns of Table 3 in terms of statistical significance and the magnitude of the coefficients. Again, we find that upward changes in risk significantly affect house prices, but downward changes do not. Since using geographical trend controls at different levels of local aggregation (1km grid and MSOA level) yield very similar results in conjunction with our preference towards the more restricted 1km grid fixed effects strategy, we adopt the coefficients from columns 1 and 2 as our main results and use them in our subsequent analysis.

5.2 Sorting

In this section we investigate the possibility that changes in radon risk information may lead to tastebased sorting of residents across properties for two main reasons. First, testing for sorting might help us to shed light on issues related to environmental justice (e.g. Banzhaf et al., 2019; Hausman and Stolper, 2021). Second, it is unclear whether our estimates above can be classified as the average willingness to pay to avoid radon of the average buyer, since it requires the strict assumption that buyers do not sort based on radon (Greenstone, 2017; Gamper-Rabindran and Timmins, 2013; Davis,

 $^{^{23}}$ We have looked at the media coverage on Radon in the United Kingdom over the 6 months before and after the changes in the 2007 map using data from the market-leading online news aggregation platform Factiva. In particular, we restricted our search to UK sources and used the keyword 'radon' in our search. We did not find evidence for meaningful changes in the media coverage on the topic of radon at and around the time of the publication of the new map in November 2007. Nevertheless, it is important to mention that there were some explicit coverage of this new map by The Times and the Daily Post, and it was also very briefly mentioned in the Sunday Sun. Overall, this exercise suggests that the radon map change was not extensively covered by the media and support the notion that the information on the new radon risk levels are mainly coming from conveyancing searches.

2004).

In Table 4 we test whether residents sort in response to changes in radon risk using census data by regressing changes in Output Area characteristics between the 2001 and 2011 Census on our main radon treatment variables (collapsed at the Output Area level), conditional on our preferred 1km postcode grid fixed effects.²⁴ The table shows that the share of people with characteristics normally correlated with higher house prices, such as education, or wealth (as proxied by owning an house), decreases in response to upward radon risk reclassification (Ioannides and Zabel, 2003; van Ham and Manley, 2009). We also find no strong evidence that the ethnic composition (as proxied by the share of white people) changes in areas that are newly affected by radon, which is interesting in light of the recent discussion around racial gaps in pollution exposure in the U.S (Tessum et al., 2019; Colmer et al., 2020). Overall these findings lead us to two important conclusions. First, radon risk falls disproportionately on lower socio-economic²⁵ groups, as they move into affected areas in search for a price discount, an example of environmental injustice that results from sorting. In fact, despite radon risk being mitigable to some extent, the risk could fall disproportionately on lower SEGs if they do not install remedial measures. Evidence from surveys show that lower SEGs are not as aware of the risks of radon (Zhang et al., 2011). Our findings are also in line with the theoretical predictions of Hausman and Stolper (2021), claiming that environmental inequities can arise even from uniform limitations to perfect information when all individuals are equally uninformed, producing disproportionate pollution exposure and welfare loss for low-income households, mostly because of sort according to known pollution. In addition, the installation of remedial measures by households in lower SEGs could be limited by more stringent budget constraints. Second, we are unable to support the assumption of no sorting as a consequence of the reclassification, and we therefore interpret our previous estimates as "local" in the sense that they reflect the difference between the WTP of the ex-ante and ex-post marginal buyers after the property have been reclassified. In other words, it is the market price of indoor air pollution risk in the housing market, under supply determined by the pre-determined housing stock and the natural experiment, while demand is determined by preferences of the population. However, since we are also interested in the average willingness to pay for avoiding radon, in Section 6 we develop a new theoretical framework to estimate this policy relevant parameter.

²⁴Note that the use of 2001 does not exactly mirror our house price results because reporting radon to home buyers became compulsory in 2002 and new radon maps were issued in November 2003 and 2007. These results here will capture the composite effect of these changes rather than just the 2007 map change, and we interpret them as about the link between radon and sorting rather than specifically about the effect of the 2007 map.

 $^{^{25}}$ We define SEGs by education which in England is highly correlated to other measures such as social class or occupation. For example, as noted earlier the correlation of share educated and share professional and managers is very high.

5.3 Robustness and Heterogeneity Tests

In this section we provide a set of robustness tests to evaluate the soundness of our main empirical result, and heterogeneity analysis across several dimensions of house and area characteristics. We begin by conducting a placebo test which examines the impact of the change in radon maps introduced in 2007 on transaction pairs in which both first and second sales happen before the publication of the 2007 map, the period between August 2002 and October 2007. This approach checks if local trends are correlated with changes in radon, including long-term trends induced by sorting. The result of this test is presented in the first column of Table 5. Reassuringly, we find no effect, which demonstrates that our treatment effect cannot be attributed to any other long-term trends that are correlated to changes in radon.

In column (2) of Table 5, we assess if our estimates are affected by measurement error, as previously discussed in Section 4. To address this issue, we take advantage of the fact that most house radon measurements in the UK are recorded by PHE (the government agency which subsidises radon measurements). Our strategy is to replicate our main analysis but restricting the focus only to areas where fewer than 1% of houses have had measurements taken by PHE. Our estimate in in column (2) is highly significant and almost identical to our baseline results, demonstrating that house radon measurement are unlikely to be a major source of bias in our estimates. This finding thus provides further reassurance that our main results are robust to mis-measurement that may arise as a consequence of a dwelling having an evaluation of radon risk independent of the publicly-available radon maps.

In column (3) of Table 5, we test whether our results are robust to using a house-level fixed-effect panel rather than a first difference estimating approach, and find that they are (Davis, 2004). Although the house fixed-effect has the advantage of using more information, we generally prefer to rely on the first difference approach as it gives us more flexibility in dealing with radon as a categorical variable and provides a clearer match to our theoretical model.

Next, we focus on houses purchased with a mortgage, since cash buyers are not strictly required to request an environmental search for their house (CON29). In this case, it is possible that the buyer would not receive information about radon risk. To test if this is affecting our results, we use a sub-sample of transactions financed by the mortgage provider Nationwide Building Society. To assess if transactions with mortgages are different from the rest of the sample, we add a dummy variable denoting having a mortgage from Nationwide and interact it with the treatment variable. The results presented in column (4) of Table 5, indicates a statistically indistinguishable but smaller effect

for mortgaged homes which suggests that our results are unlikely to be biased by mis-measuring the information that cash buyers receive.

We report further robustness checks in the Appendix. We start by using an alternative identification strategy based on the spatial boundary discontinuity design (BDD) akin to Gibbons et al. (2013) exploiting boundaries of radon grid tiles. As lease term is unobserved yet critical to house value in the cross-section, we focus on freehold transactions within 300m of the boundary between radon affected (category 2) and radon free (category 1) areas.²⁶²⁷²⁸ To further strengthen the identification, we also control for third-order polynomials in distance from the boundary, which account for possible spatial trends in house prices closer to boundaries.²⁹ The results of our BDD regressions are presented in Table A3 in the Appendix. The first column shows cross sectional results across the BDD sample with distance to the boundary polynomials as well as TTWA and time period fixed effects. In the second column we add property characteristics to the sample to ensure that our results are not biased by changes in characteristics of the housing stock at the boundary. In the third column, we add Output Area characteristics from the 2001 Census as control variables (the same as in our sorting table 4) to control for cross sectional differences in housing market characteristics. Finally, we extend our bandwidth to 500m and show that our key coefficient remains close to our other results. Overall, our BDD analysis yields results which are very similar to our first-difference estimates. However, the results are clearly less precisely estimated than our main results in Table 3. This seems logical given the smaller sample size, less precise controls for unobserved property characteristics and the fact that not all sellers were informed about radon when they purchased their properties. 30

²⁹Formally, we estimate the treatment effect given by β in the boundary sample in the following equation:

$$ln(P_{it}) = \beta RA_{it} + D_i + T_t + TTWA_i + \epsilon_{it}$$

$$\tag{7}$$

where P_{it} denotes the transaction price of property *i* in period (quarter) *t*, RA_{it} is a dummy variable that equals one if the property is on the side of the boundary that is affected by radon, D_i is a vector of distance to the boundary polynomials up to the third order, T_t is a time period (quarter) fixed-effect, $TTWA_i$ is the Travel to Work Area local labour market fixed-effect, and ϵ is an idiosyncratic error term.

 $^{^{26}}$ Since some areas consist of only one tile, we use 300m as a bandwidth of roughly one third of a tile's length/height, but get similar results with other bandwidths.

 $^{^{27}}$ We use the $1km^2$ gird, in locations/periods that did not have access to the 25x25 meter grid map, and transacted between 2007 and 2012.

²⁸Freehold of a property means that the buyer owns it outright in perpetuity, including the land it is built on. This is the most common mode of ownership of real property in the UK, the Commonwealth and most European countries, in contrast to leasehold, in which the property reverts to the owner of the land after the lease period expires.

³⁰Compared to our first-difference specification, BDD has some notable shortcomings. First, it focuses on a smaller sample, as we can only use transactions close to boundaries. Second, it cannot show if the estimated effect is attributable to increasing or decreasing radon risk. Third, it is only suitable for large grids $(1km^2 \text{ and } 5km^2)$ and cannot be used with the most accurate map (25 meter square). Nonetheless, a cross-sectional BDD approach is very useful as a robustness

In our last set of robustness tests we explore the assumption that prices react to radon information given to buyers by Local Authorities through the CON29 form. More specifically we leverage that we know which version of the 2007 radon map - the 25x25 meter or the $1km^2$ grid - is being used by a LA to provide radon risk information at the time of the second sale, and we run tests to check that it is the information which is indeed driving house price changes. In column 1 of Table A4 we first separate out the treatment variable for the two different versions of the map and estimate the effect for each group.³¹ We find that the impact of radon reclassification to "radon affected" is the same regardless of which version of the map is being used to provide this information. In column 2 we restrict attention to sales where CON29 is based on the 25 meter square map and test if a radon treatment defined using the $1km^2$ grid information affects prices. We find that it does not. However, more reassuringly, when information on radon changes from both $1km^2$ and 25x25 meter grids are included in the same regression in column 3, only the latter matters for transactions that have access to it. In columns 4 and 5 we conduct similar tests to those performed in columns 2 and 3 but this time we restrict attention to sales where CON29 is based on the $1km^2$ grid map and test if a radon treatment defined using the 25x25 meter grid information affects prices. Again, we find that only the estimate for the $1km^2$ grid is statistically significant. This provides further convincing evidence that the effects we estimate stem from information transmitted in the CON29 form.

A related issue is that the accuracy and consistency of radon information buyers receive could influence the way that information about risk translates into prices. One source of potential inaccuracy is Home Information Packs (HIPs). These were in place throughout the majority of our second sale sample time frame and the limited evidence available suggests that HIPs may have contained at least some degree of inaccurate information (although not necessarily about radon). As our main results compare homes that stay radon-unaffected with those that become affected under the map change, if we assume that HIPs contained radon information from the earlier map then buyers of newly radonaffected homes would get conflicting information from the HIP and the later results from their own solicitor (via the CON29 form). If so our estimate may underestimate the effect of radon risk to the extent that buyers become mistrustful of the radon information in the CON29 form and price this in. To assess whether this is a threat to our identification in Table A5 we report additional regression specifications. Column 1 replicates our main estimate which includes pairs with second sales in the

check of our main strategy, as it is based on different identification assumptions. Specifically, it assumes that while the information about being affected by radon is discontinuous at the boundary of a grid tile, all other determinants of house prices are continuous.

³¹Note that we use $1 \rightarrow 2$ radon treatment for this test. Using the $1 \rightarrow Any$ treatment yields qualitatively similar results, but these are estimated with less precision.

interval 2008-2011 inclusive. In column 2 we alter the sample to those pairs where the second sale takes place in 2011 or 2012, which should eliminate any effect of HIP misinformation as HIPs were abolished in 2010. We find the coefficient is essentially unchanged to column 1. A second test is underpinned by our belief that potential inaccuracies in radon information in general (either in HIPs or indeed local authority searches) would be very likely to be reduced after the 2007 map became fully embedded in the system. We therefore find it reassuring in column 3 that estimate is identical when we exclude second sales in the first year following the map change. Overall these tests lead us to conclude that buyers' mistrust of radon information, while potentially an important qualifier, is in practice not a meaningful factor in this specific context.

Finally, we investigate heterogeneity using models that employ our preferred 1km grid fixed-effects. In Table 6 our approach is to estimate the baseline model in column 2 of Table 3, but including separate radon treatment terms for different groups of observations (e.g. flats and houses) while simultaneously controlling for time effects for the groups, as well as the interaction between the group variable and a new build indicator. In column (1) we examine price effects for flats and houses, where we find the effects on houses to be very considerably larger and the effects on flats to be insignificant. This likely reflects heterogeneity with respect to distance to the ground because in column (2) we find very large negative effects of radon for basement flats. In column (3) and (4) we find that effects are generally slightly larger in London and in urban conurbations than elsewhere. While these results may be suggestive of heterogeneity, it is important to note that we do not have sufficient evidence to reject the null that the coefficients are equal in all columns, except for the first where we find the difference between flats and houses is statistically significant (F-test *p*-value = 0.04).

The heterogeneity we observe in these latter results could plausibly reflect differences in risk updating. Our priors are that more educated and wealthier residents may update their beliefs more than other residents. We lack the micro-data necessary to make strong claims in this respect. However, when we estimate separate effects for different quantiles of neighbourhood education and home-ownership, we obtain mixed results as shown in Figure 4. We are therefore unable to draw strong conclusions about whether there is differential updating of beliefs about radon risk in this setting.

6 Identifying WTP

As different households have different preferences, average WTP is not observed directly in housing transactions data as houses are purchased by the highest bidders (see Section A3 in the Appendix for details). However, we develop a strategy that allows us to estimate WTP from the impact of radon

on both transaction prices and the number of transactions that occur at those prices. The idea is that if the transaction price is the market-clearing price, the probability of a transaction at that price can help us to reveal the WTP of an average treated owner. We briefly summarize the idea below, and section A3 explains in more detail this idea with a simple theoretical model and translates it into an empirical strategy.

At the very basic level, a house is sold if the value it delivers to the owner is lower than the value it delivers to the highest bidder. Both of these values are determined by the same household problem (outlined in section A3), and the results differ due to different budgets and preferences. Based on the evidence that our treatment is orthogonal to characteristics of owners, we can assume that the distribution of preferences of the treated owners is random (reflecting different characteristics of households). At the same time, the sorting mechanism suggests that the composition of preferences of the highest bidders is likely to be endogenous to prices and the corresponding characteristics of properties in the housing stock. For example, properties affected by radon attract buyers who do not care about radon. The discount they can negotiate on radon affected houses depends on how much households who want to avoid radon are willing to pay to avoid it. Owners of affected houses can choose to sell at market prices or to tolerate radon risk. The likelihood of an average owner deciding to sell after being informed that their house has been reclassified as affected by radon, and that the market value of their house has decreased by the average radon discount, reveals their willingness to pay to avoid radon.

In Section A2, we find that the probability of a transaction increases by only 0.0026%. This is a very small change over the mean transaction probability of 0.087% before the treatment. Not surprisingly, after plugging it into our model we conclude that the impact of sorting is very small, and that the hedonic estimate of 0.86% (or £1,760) is very close to the average WTP of 0.89% (or £1,816). It is worth noting that the result is driven by the fact that the change in transaction probability is very small, and that parameters of our model make very little difference. Importantly, the fact that the change in transaction probability is small does not imply that households do not sort, but simply that the impact this has on prices is negligible. Thus, the estimated discount of prices and increase in transactions seem to be mainly driven by the WTP to avoid radon-related health risks rather than sorting of different socio-economic groups into the neighbourhood.

7 Conclusion

We are among the first to provide evidence on the economic effects of indoor air pollution. By focusing on radon and the housing market, we are able to address many of the key challenges in the existing air pollution literature and provide compelling evidence that market participants react to information about indoor air pollution related to radon. Our results show that prices of houses affected by exogenous increases in radon risk decrease significantly. Our preferred specifications place the magnitude of the estimate at around 0.8%. We also find that changes in radon risk cause sorting and corresponding changes in characteristics of residents of the affected areas. Specifically, we show that residents from higher SEGs move out of places that are newly affected by radon, and that people from lower SEGs move in. These results thus highlight how sorting leads to the disproportionate exposure of lower SEGs to indoor air pollution. Although beyond the scope of our paper, we note that this environmental injustice may be exacerbated if individuals in these SEGs are less able to adopt radon remedial measures, for example because of credit or informational constraints.³²

Our results prove to be robust to multiple tests, but they are still limited by the institutional setting and data availability. Most notably, while we find no evidence that reclassifying a house from a radonaffected to a radon-free category increases prices, we are unable to conclusively determine whether this is due to sellers being unaware of changes to radon levels (unlike the buyers), or due to a relatively small sample size. For similar reason, we are unable to exploit changes in the magnitude of radon risk in affected houses to understand how households react to changes in risk more generally. One particular question that we are unable to definitively answer is whether we can interpret our estimates as directly reflecting households valuation of the health risks of radon exposure.³³ This is because in our context a series of issues, including the ability of household to install remedial measures to mitigate risks and uncertainty around radon risk information quality, may drive a wedge between our estimates and households valuations.³⁴ That said, we do find it instructive to compare our estimates with both the implied value of health risks from radon, as well as the financial costs of installing radon remediation measures. Using the results discussed above, our estimated coefficient for moving up from

³²This seems plausible because surveys suggest the main reason for households not remediating radon is cost. See Appendix D of the report of the Independent Advisory Group on Ionising Radiation https://assets.publishing. service.gov.uk/government/uploads/system/uploads/attachment_data/file/335102/RCE-11_for_website.pdf.

³³Related work in the literature (Davis, 2004; Sunstein and Zeckhauser, 2011; Dessaint and Matray, 2017) has typically found that market participants seemingly overreact to health risk.

³⁴Note that our robustness checks do provide some reassurance that information inaccuracies do not drive our findings. While we cannot rule out the possibility that buyers may anticipate regulations that oblige them to incur full remediation costs before re-selling the house, we are not aware this issue has ever been proposed or mentioned, and moreover the regulation of similar risks (like flood risk) has been consistently based on information disclosure rather than mandatory remediation in our setting.

category 1 to category 2 in column 1 panel A of Table 3 implies a change in house prices of roughly $\pounds 1,760$. Based on a necessarily assumption heavy approach, we estimate that moving from category 1 to category 2 would yield an increase in risk with a present value of between $\pounds 800$ and $\pounds 1,400$ per household on average which is in the same ballpark.³⁵ A further relevant comparison is to the typical cost of installing radon mitigation systems. These are usually between $\pounds 200$ and $\pounds 2000$ according to Public Health England, and hence are also very similar to our main estimates. Whether remediation costs bound our estimated price discounts from above or below is a question we leave to future work.

We also provide new evidence on how the housing market reacts to information about environmental risks. We show that mandatory information provision has a strong effect on household choices and that, at least in the case of radon, prices react to the information about the risk even when the actual level of risk is the same. Therefore, alternative policy initiatives could be more efficient in order to provide health risk information to the population, while preventing the depression of the housing market value. For instance, CON29 forms could provide more detailed information, in particular distinguishing between the different radon categories and relative risk levels, and also disclosing all radon measurements done at the property level, especially in radon-affected areas. This would provide more advice and support to homeowners, and might also increase remediation rates. Following the suggestions made by Gray et al. (2009), an effective intervention would require the combination of two policies. First, extending the existing policy requiring measurement and basic preventive measures in all new residential properties from high radon level areas to the entire UK would be highly cost effective, reducing exposure levels and providing precise information to buyers. Secondly, a campaign of mass testing to precisely measure the exposure to radon for all residential properties in high radon level areas would be highly beneficial. This approach would solve once for all the information mismatch problem in affected areas at a limited cost of around £15 per test. Property-level low radon concentrations would be favourably reflected in the price of houses located in radon-affected areas, while documented high levels could increase the likelihood of remediation measures installed by homeowners.

³⁵To compute this we use the cumulative risks of lung cancer at different radon concentrations (category levels) by smoking status, from Table 4.3 of the Health Protection Agency report "Radon and Public Health: Report of the independent Advisory Group on Ionising Radiation" https://assets.publishing.service.gov.uk/government/uploads/system/uploads/attachment_data/file/335102/RCE-11_for_website.pdf. We calculate the implied monetary values of lung cancer risk in each using a value of a statistical life of £5 million. This VSL is of course arbitrary, but it is within the typical range used in the US. The comparison of a transformed VSL to the reduced-form price effect relies on mean risk and variance as sufficient statistics (Cordoba and Ripoll, 2016). To assign risk values to households we assume households contain 2.3 people, that 10% are smokers (according to the UK Office of National Statistics, 14.7% of people aged 18 years and above are smokers https://www.ons.gov.uk/peoplepopulationandcommunity/healthandsocialcare/healthandlifeexpectancies/bulletins/adultsmokinghabitsingreatbritain/2018), and that adults represent 70% of the population.

Overall, our analysis highlights the importance of considering the extent and distribution of indoor air pollution costs alongside those of ambient air pollutants. Given the significant amount of time that we spend indoors, and the relatively little policy and research attention to this subject, we argue that much more focus should be directed towards this issue in future work.

References

- AGIR (2009). Radon and Public Health. Report of the independent Advisory Group on Ionising Radiation. Documents of the Health Protection Agency. Radiation, Chemical and Environmental Hazards RCE-11, Health Protection Agency.
- Bakkensen, L. A. and Barrage, L. (2017). Flood Risk Belief Heterogeneity and Coastal Home Price Dynamics: Going Under Water? NBER Working Papers 23854, National Bureau of Economic Research, Inc.
- Bakkensen, L. A. and Ma, L. (2020). Sorting over flood risk and implications for policy reform. Journal of Environmental Economics and Management, 104:102362.
- Banzhaf, S., Ma, L., and Timmins, C. (2019). Environmental justice: The economics of race, place, and pollution. *Journal of Economic Perspectives*, 33(1):185–208.
- Barwick, P. J., Li, S., Lin, L., and Zou, E. (2019). From fog to smog: The value of pollution information. Technical report, National Bureau of Economic Research.
- Bayer, P., Ferreira, F., and McMillan, R. (2007). A Unified Framework for Measuring Preferences for Schools and Neighborhoods. *Journal of Political Economy*, 115(4):588–638.
- Besley, T., Meads, N., and Surico, P. (2014). The incidence of transaction taxes: Evidence from a stamp duty holiday. *Journal of Public Economics*, 119(C):61–70.
- Bosker, M., Garretsen, H., Marlet, G., and van Woerkens, C. (2018). Nether Lands: Evidence on the Price and Perception of Rare Natural Disasters. *Journal of the European Economic Association*, 17(2):413–453.
- Cannaday, R. E., Munneke, H. J., and Yang, T. T. (2005). A multivariate repeat-sales model for estimating house price indices. *Journal of Urban Economics*, 57(2):320–342.
- Chay, K. Y. and Greenstone, M. (2005). Does Air Quality Matter? Evidence from the Housing Market. Journal of Political Economy, 113(2):376–424.
- Colmer, J., Hardman, I., Shimshack, J., and Voorheis, J. (2020). Disparities in pm2. 5 air pollution in the united states. *Science*, 369(6503):575–578.
- Cordoba, J. C. and Ripoll, M. (2016). Risk Aversion and the Value of Life. The Review of Economic Studies, 84(4):1472–1509.

- Currie, J., Davis, L., Greenstone, M., and Walker, R. (2015). Environmental Health Risks and Housing Values: Evidence from 1,600 Toxic Plant Openings and Closings. *American Economic Review*, 105(2):678–709.
- Cutter, W. B. and Neidell, M. (2009). Voluntary information programs and environmental regulation: Evidence from 'spare the air'. *Journal of Environmental Economics and management*, 58(3):253–265.
- Darby, S., Hill, D., Deo, H., Auvinen, A., Barros-Dios, J. M., Baysson, H., Bochicchio, F., Falk, R., Farchi, S., Figueiras, A., Hakama, M., Heid, I., Hunter, N., Kreienbrock, L., Kreuzer, M., Lagarde, F., Mäkeläinen, I., Muirhead, C., Oberaigner, W., Pershagen, G., Ruosteenoja, E., Rosario, A. S., Tirmarche, M., Tomášek, L., Whitley, E., Wichmann, H.-E., and Doll, R. (2006). Residential radon and lung cancer—detailed results of a collaborative analysis of individual data on 7148 persons with lung cancer and 14 208 persons without lung cancer from 13 epidemiologic studies in europe. *Scandinavian Journal of Work, Environment & Health*, 32:1–84.
- Davis, L. W. (2004). The effect of health risk on housing values: Evidence from a cancer cluster. American Economic Review, 94(5):1693–1704.
- Dessaint, O. and Matray, A. (2017). Do managers overreact to salient risks? Evidence from hurricane strikes. *Journal of Financial Economics*, 126(1):97–121.
- Duflo, E., Greenstone, M., and Hanna, R. (2008). Indoor air pollution, health and economic well-being. S.A.P.I.EN.S. - Surveys and Perspectives Integrating Environment and Society.
- Ekeland, I., Heckman, J. J., and Nesheim, L. (2002). Identifying hedonic models. American Economic Review, 92(2):304–309.
- Gamper-Rabindran, S. and Timmins, C. (2013). Does cleanup of hazardous waste sites raise housing values? evidence of spatially localized benefits. *Journal of Environmental Economics and Management*, 65(3):345–360.
- Gibbons, S., Machin, S., and Silva, O. (2013). Valuing school quality using boundary discontinuities. Journal of Urban Economics, 75:15–28.
- Gilraine, M. (2019). Air Filters, Pollution and Student Achievement. Technical Report 188, Annenberg Institute at Brown University.
- Gray, A., Read, S., McGale, P., and Darby, S. (2009). Lung cancer deaths from indoor radon and the cost effectiveness and potential of policies to reduce them. *BMJ*, 338.

- Greenstone, M. (2017). The Continuing Impact of Sherwin Rosen's "Hedonic Prices and Implicit Markets: Product Differentiation in Pure Competition". *Journal of Political Economy*, 125(6):1891– 1902.
- Gruber, V., Bossew, P., Cort, M. D., and Tollefsen, T. (2013). The European map of the geogenic radon potential. *Journal of Radiological Protection*, 33(1):51–60.
- Hastings, J. S. and Weinstein, J. M. (2008). Information, school choice, and academic achievement: Evidence from two experiments. *The Quarterly journal of economics*, 123(4):1373–1414.
- Hausman, C. and Stolper, S. (2021). Inequality, information failures, and air pollution. Journal of Environmental Economics and Management, 110:102552.
- Holgate, S. T. (2017). 'every breath we take: the lifelong impact of air pollution'-a call for action. *Clinical Medicine*, 17(1):8.
- Howe, G. R., Nair, R. C., Newcombe, H. B., Miller, A. B., and Abbatt, J. D. (1986). Lung Cancer Mortality (1950–80) in Relation to Radon Daughter Exposure in a Cohort of Workers at the Eldorado Beaverlodge Uranium Mine. JNCI: Journal of the National Cancer Institute, 77(2):357–362.
- Howe, J. and Shreedhar, G. (2020). Do Healthy Home Checklists Nudge People to Test Their Home for Radon?: Evidence from a Randomised Control Trial. Grantham Research Institute on Climate Change and the Environment.
- Ioannides, Y. M. and Zabel, J. E. (2003). Neighbourhood effects and housing demand. Journal of Applied Econometrics, 18(5):563–584.
- Ito, K. and Zhang, S. (2020). Willingness to pay for clean air: Evidence from air purifier markets in china. Journal of Political Economy, 128(5):1627–1672.
- Jessoe, K. and Rapson, D. (2014). Knowledge is (less) power: Experimental evidence from residential energy use. American Economic Review, 104(4):1417–38.
- Kask, S. B. and Maani, S. A. (1992). Uncertainty, information, and hedonic pricing. Land Economics, 68(2):170–184.
- Klepeis, N., Nelson, W., Ott, W. R., Robinson, J. P., Tsang, A. M., Switzer, P., Behar, J. V., Hern, S. C., and Engelmann, W. H. (2001). The National Human Activity Pattern Survey (NHAPS): a resource for assessing exposure to environmental pollutants. *Journal of Exposure Science and Environmental Epidemiology*, 11:231–252.

- Krewski, D., Lubin, J. H., Zielinski, J. M., Alavanja, M., Catalan, V. S., Field, R. W., Klotz, J. B.,
 Létourneau, E. G., Lynch, C. F., Lyon, J. I., Sandler, D. P., Schoenberg, J. B., Steck, D. J., Stolwijk,
 J. A., Weinberg, C., and Wilcox, H. B. (2005). Residential Radon and Risk of Lung Cancer: A
 Combined Analysis of 7 North American Case-Control Studies. *Epidemiology*, 16(2):137–145.
- Künn, S., Palacios, J., and Pestel, N. (2019). Indoor air quality and cognitive performance. Technical report, IZA Discussion Paper.
- Lubin, J. H., Wang, Z. Y., Boice Jr, J. D., Xu, Z. Y., Blot, W. J., De Wang, L., and Kleinerman, R. A. (2004). Risk of lung cancer and residential radon in China: Pooled results of two studies. *International Journal of Cancer*, 109(1):132–137.
- McCluskey, J. J. and Rausser, G. C. (2003). Stigmatized asset value: Is it temporary or long-term? The Review of Economics and Statistics, 85(2):276–285.
- Miles, J., Appleton, J., Rees, D., Green, B., Adlam, K., and Myers, A. (2007). *Indicative atlas of radon in England and Wales*. Health Protection Agency Chilton, Oxon.
- Moulton, J. G., Sanders, N. J., and Wentland, S. A. (2018). Toxic assets: How the housing market responds to environmental information shocks. Technical report, Department of Economics Working Papers, College of William and Mary.
- Naoi, M., Seko, M., and Sumita, K. (2009). Earthquake risk and housing prices in Japan: Evidence before and after massive earthquakes. *Regional Science and Urban Economics*, 39(6):658–669.
- NRPB (1996). Radon Affected Areas: England. Doc NRPB 7(2): 1-9, NRPB.
- PHE (2018). UK National Radon Action Plan. Public Health England.
- Pope, J. C. (2008). Buyer information and the hedonic: the impact of a seller disclosure on the implicit price for airport noise. *Journal of Urban Economics*, 63(2):498–516.
- Radford, E. P. and Renard, K. G. S. C. (1984). Lung Cancer in Swedish Iron Miners Exposed to Low Doses of Radon Daughters. New England Journal of Medicine, 310(23):1485–1494.
- Roth, S. (2019). The contemporaneous effect of indoor air pollution on cognitive performance: Evidence from the UK. In *mimeo*.
- Smith, V. K. and Johnson, F. R. (1988). How do Risk Perceptions Respond to Information? The Case of Radon. The Review of Economics and Statistics, 70(1):1–8.
- Stafford, T. M. (2015). Indoor air quality and academic performance. Journal of Environmental Economics and Management, 70:34–50.

- Sunstein, C. R. and Zeckhauser, R. (2011). Overreaction to Fearsome Risks. Environmental and Resource Economics, 48:435–449.
- Tessum, C. W., Apte, J. S., Goodkind, A. L., Muller, N. Z., Mullins, K. A., Paolella, D. A., Polasky, S., Springer, N. P., Thakrar, S. K., Marshall, J. D., et al. (2019). Inequity in consumption of goods and services adds to racial–ethnic disparities in air pollution exposure. *Proceedings of the National Academy of Sciences*, 116(13):6001–6006.
- Tirmarche, M., Raphalen, A., Allin, F., Chameaud, J., and Bredon, P. (1993). Mortality of a cohort of French uranium miners exposed to relatively low radon concentrations. *British Journal of Cancer*, 67:1090–1097.
- US-EPA (2013). EPA map of radon zones.
- van Ham, M. and Manley, D. (2009). The effect of neighbourhood housing tenure mix on labour market outcomes: A longitudinal investigation of neighbourhood effects. *Journal of Economic Geography*, 10(2):257–282.
- WHO (2009). WHO handbook on indoor radon: A public health perspective. Technical report, World Health Organization.
- Wichman, C. J. (2017). Information provision and consumer behavior: A natural experiment in billing frequency. Journal of Public Economics, 152:13–33.
- Zhang, W., Chow, Y., Meara, J., and Green, M. (2011). Evaluation and equity audit of the domestic radon programme in england. *Health Policy*, 102(1):81–88.
- Zielinski, J. M., Carr, Z., Krewski, D., and Repacholi, M. (2006). World Health Organization's International Radon Project. Journal of Toxicology and Environmental Health, Part A, 69(7-8):759–769. PMID: 16608837.
- Zivin, J. G. and Neidell, M. (2009). Days of haze: Environmental information disclosure and intertemporal avoidance behavior. *Journal of Environmental Economics and Management*, 58(2):119–128.

8 Tables and Figures

Figure 1: Radon Maps



Notes: The maps give the level of radon risk from 1 to 5 in the first map and from 1 to 6 in the other two with darker blue corresponding to higher levels of radon.

Figure 2: Areas reclassified between "radon-free" and "radon-affected" from the 2003 to the 2007 maps.



Notes: Areas in black are reclassified from "radon-free" in 2003 to "radon-affected" in 2007. Areas in grey are reclassified from "radon-affected" to "radon-free". Areas in white have not been affected by the reclassification.

	Radon 2nd sale						
Radon 1st sale	1(Radon-Free)	2	3	4	5	6	Total
1(Radon-Free)	411,262	81,918	12,826	9,036	$3,\!950$	455	519,447
2	20,012	$16,\!584$	5,999	$6,\!125$	$4,\!520$	496	53,736
3	1,826	$2,\!154$	1,714	$2,\!537$	$2,\!857$	652	11,740
4	587	411	416	874	$3,\!561$	$1,\!112$	6,961
5	139	187	133	688	3,404	3,801	8,352
6	0	1	0	2	161	$1,\!255$	1,419
Total	433,826	$101,\!255$	$21,\!088$	$19,\!262$	$18,\!453$	7,771	$601,\!655$

Table 1: Summary statistics: Repeated transactions with changing radon levels.

Notes: the table gives the number of repeated transactions for the estimation sample, which includes only those homes with a first sale in the period Aug 2002 to October 2007 and a second sale in the period January 2008 to December 2011. Data on transactions comes from Land Registry and on radon levels from PHE.

		(1)	(2)	(3)	(4)	(5)
		Radon-Free	Radon-Affected	$\Delta radon=0$	$\Delta radon>0$	$\Delta radon < 0$
Price	Mean	$198,\!846$	187,706			
	St. Dev.	194,031	$161,\!697$			
Change in Price	Mean			$35{,}542$	$23,\!965$	$23,\!328$
	St. Dev.			119,963	61,005	68,215
N		7,095,851	1,993,193	435,093	139,845	26,717

Table 2: Summary statistics: Average prices of house prices and changes in radon.

Notes: the table gives the number of individual transactions in columns 1-2 and repeated transactions with no change in radon risk, upwards change, and downward change in columns 3-5 respectively. The data is the estimation period which means that only first transactions after July 2002 and second transactions between 2008 and 2012 are included. Data on transactions comes from Land Registry and on radon levels from PHE. Radon-Free indicates radon category 1; Radon-Affected indicates all higher categories.

Figure 3: Average price for radon-affected reclassified and unaffected properties before and after the reclassification.



Notes: the figure shows the evolution of prices, comparing properties that were reclassified from radon free to radon-affected areas with unaffected properties before and after the reclassification in October 2007.

	(1)	(2)	(3)	(4)
	Grid FX		MSOA FX	
	$X{=}\{2\}$	$X{=}\{All\}$	$X = \{2\}$	$X{=}\{All\}$
Panel A: $1 \to X$	-0.0086+	-0.0076+	-0.0078+	-0.0063+
	(0.0022)	(0.0020)	(0.0020)	(0.0018)
N	449563	473873	449763	473134
R^2	0.498	0.498	0.414	0.413
Panel B: $1 \to X$	0.0038	0.0032	-0.0001	-0.0023
	(0.0067)	(0.0064)	(0.0059)	(0.0056)
N	26449	32387	26457	32364
R^2	0.506	0.519	0.416	0.428

Table 3: Radon Risk Reclassification and House Prices

Notes: Standard errors in parentheses, clustered at 1km radon grid. * p < 0.1, ** p < 0.05, *** p < 0.01, $\dagger p < 0.001$. All regressions include interactions of years of sale with area (postcode grid or MSOA) indicators as well as quarter of first and second sale fixed effects. Panel A tests for upwards risk moves, using samples composed of sales that stay in category 1 (radon free) and houses that move from category 1 to radon risk categories. In columns 1 and 3, we examine moves from category 1 to the lowest category classified as affected by radon (category 2). Columns 2 and 4 include houses that start in category 1 but move to any higher risk categories. Panel B repeats the same process for downwards risk moves.

	(1)	(2)	(3)	(4)	(5)
	A Level	Degree	Own house	Own house	Ethnic
	or higher	Education	Outright	Any	White
$1 \rightarrow Any$	-0.0042^{\dagger}	-0.0020**	0.0017	-0.0042***	-0.0010
	(0.0011)	(0.0009)	(0.0011)	(0.0013)	(0.0011)
$Any \rightarrow 1$	-0.0007 (0.0028)	-0.0005 (0.024)	0.0054 (0.0038)	-0.0014 (0.050)	-0.0008 (0.0025)
N	134107	134107	134107	134107	134107
R^2	0.375	0.308	0.361	0.343	0.653

Table 4: Radon Risk and Sorting

Notes: Standard errors in parentheses, clustered at the 1km radon grid level . ** p < 0.05, *** p < 0.01, $\dagger p < 0.001$. The table reports results from regressing changes in Output Area characteristics between the 2001 and 2011 Census on our main radon treatment variables (collapsed at the Output Area level), conditional on our preferred 1km postcode grid fixed effects. The first row presents the results of moving from "radon-free" to "radon-affected" categories. The second row shows the results of moving from "radon-affected" categories to the "radon-free" risk category. Column headings denote micro-neighbourhood characteristics as shares of population; (1) with A levels or higher as the highest level of education, (2) with Degree education as the highest level of education, (3) owning their houses outright, (4) owning their houses outright or with a mortgage, and (5) identifying as white.

	(1)	(2)	(3)	(4)
	Placebo	Radon Tests	Panel House FX	Mortgage
	$\Delta ln(price)$	$\Delta ln(price)$	ln(price)	$\Delta ln(price)$
$1 \rightarrow Any$	-0.0000	-0.0082^{\dagger}	-0.0049***	-0.0086^{\dagger}
	(0.0018)	(0.0021)	(0.0019)	(0.0022)
$1 \rightarrow Any$ with Mortgage				-0.0004 (0.0090)
N	566845	450994	2438192	449563
R^2	0.400	0.498	0.966	0.498

Table 5: Placebo and Robustness Tests

Notes: Standard errors in parentheses clustered at the 1km radon grid. * p < 0.1, ** p < 0.05, *** p < 0.01, †p < 0.001. All results in this table are based on an upward move from "radon-free" to "radon-affected" categories. The results in column 1 are based on pairs of transactions that occurred before the 2007 map change. The placebo change in radon applied to those pairs is based on the future radon change the house subsequently experienced. Column 2 gives results of a reclassification from no radon to any other category (again equivalent to table 3 in areas where no more than 1% of houses have been tested for radon risk by PHE. Column 3 reports results from a panel OLS regression with property fixed-effects. In column 4 we include a $1 \rightarrow 2$ with Mortgage dummy variable that equals one if the transaction is funded by a Nationwide mortgage and receives the treatment.

	(1)	(2)	(3)	(4)
	Flats	Basement	London	Conurbations
	$\Delta ln(price)$	$\Delta ln(price)$	$\Delta ln(price)$	$\Delta ln(price)$
$1 \to Any \times X = 0$	-0.0079+	-0.0075+	-0.0071***	-0.0058**
	(0.0021)	(0.0020)	(0.0023)	(0.0026)
$1 \to Any \times X = 1$	-0.0022	-0.0304*	-0.0089**	-0.0095***
	(0.0029)	(0.0166)	(0.0039)	(0.0031)
N	473873	473873	473873	473873
R^2	0.50	0.49	0.49	0.49
F test p -value	0.04	0.16	0.69	0.37

 Table 6: Heterogeneity Results

Notes: Standard errors in parentheses clustered at the 1km radon grid. * p < 0.1, ** p < 0.05, *** p < 0.01, $\dagger p < 0.001$. The table estimates our baseline model as in column 2 of Table 3 but includes separate radon treatment terms for different groups of observations while simultaneously controlling for time effects for the groups, as well as the interaction between the group variable and a new build indicator. In columns (1-4) we examine price effects on flats and houses, basement flats, and properties in London and urban conurbations respectively. The F-tests examine the null that the coefficients are equal.



Figure 4: Effects of radon by neighbourhood education share and home ownership share.

Notes: The figure estimate separate effects for different quantiles of neighbourhood education and home ownership.